Author’s response to reviews

Title: Anemia in People on Second Line Antiretroviral Treatment in Lilongwe, Malawi: a Cross-Sectional Study

Authors:

McNeil Ngongondo (mngongondo@unclilongwe.org)
Nora Rosenberg (nrosenbe@email.unc.edu)
Christopher Stanley (stanleychristopher1@yahoo.com)
Robertino Lim (robertinolim@gmail.com)
Dennis Ongubo (dongubo@tulane.edu)
Richard Broadhurst (richard_broadhurst@med.unc.edu)
Colin Speight (colinspeight88@yahoo.co.uk)
Robert Flick (robbie.flick@gmail.com)
Petros Tembo (ptembo@lighthouse.org.mw)
Mina Hosseinpour (mina_hosseinipour@med.unc.edu)

Version: 2 Date: 19 Dec 2017

Author’s response to reviews:

Dear Editor,

we have made the following revisions:

1) Please move the "List of abbreviations" and "Declaration" section to the appropriate location after "Conclusions" (https://bmcinfectdis.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article)

We have made this change putting the sections to the appropriate locations

2) At this stage, please submit a final, clean copy of your manuscript without tracked changes or highlights. Please ensure "track changes" are turned off before uploading your final manuscript.

we are submitting a final clean copy of the manuscript.
thank you,

McNeil Ngongondo